Author's response to reviews

Title: The prognostic significance of age in operated and non-operated colorectal cancer

Authors:

Jing Li (lijyz@sohu.com)
Zhu Wang (26574021@qq.com)
Xin Yuan (1499733865@qq.com)
Lichun Xu (yxyxlc@aliyun.com)
Jiandong Tong (tongjiandong@cso.org.cn)

Version: 5 Date: 23 January 2015

Author's response to reviews:

Dear editors,

Thank you for your letter concerning our manuscript entitled “The prognostic significance of age in operated and non-operated colorectal cancer (MS: 1665488313138556)”. We have studied your and reviewers comments carefully and have made correction which we hope meet with approval. Revised portion are marked in red in the paper. The following revisions have been made point-by-point based on the suggestions of your and reviewers.

Q1. Line 94- 95 -Please clarify who permitted access to data from the SEER database. Please include the name of the body which permitted data access.
Response: Done. Our permission was authorized by National Cancer Institute, USA.

Q2. Please clearly title your conclusion section.
Response: Done. We have added the title of Conclusions.

Q3. We note some irregularities concerning authorship. In your author contributions you state that Wang Zhu wrote your manuscript, however you have not included Wang Zhu as an author. Please can you clarify why Wang Zhu is not attributed as an author of your manuscript?
Response: Zhu Wang is one of authors in our manuscript. Because of my careless, I wrote his name as Wang Zhu. I have made the change.

We appreciate for Editors warm work earnestly, and hope that the correction will meet with approval.

Once again, thank you very much for your kind work.

Yours sincerely,

Jing Li

E-mail: lijyz@sohu.com